Author's response to reviews

Title: What Data is Important? Clinical Data Needs in the Neonatal Intensive Care Unit Electronic Medical Record

Authors:

Marc A Ellsworth (ellsworth.marc@mayo.edu)
Tara R Lang (lang.tara@mayo.edu)
Brian W Pickering (pickering.brian@mayo.edu)
Vitaly Herasevich (herasevich.vitaly@mayo.edu)

Version: 4 Date: 16 June 2014

Author's response to reviews: see over
Marc Ellsworth, MD  
Department of Pediatric and Adolescent Medicine  
Division of Neonatal and Perinatal Medicine  
Mayo Clinic  
200 1st Street SW  
Rochester, MN 55905  
ellsworth.marc@mayo.edu

June 16, 2014

Giulia Mangiameli, BioMed Central

Dear Editorial Board,

Thank you for allowing us to resubmit our manuscript "What data is important? Clinical data needs in the neonatal intensive care unit electronic medical record," for your review.

The following changes have been made to the manuscript:

1) The name of the IRB, with study reference number, has been added to the Methods section.
2) Page and line numbers have been added.
3) The Competing Interests section has been updated to detail that no funding was obtained for the study.

Once again, thank you for the opportunity to resubmit our manuscript. We hope that it is seriously considered for publication in *BMC Medical Informatics and Decision Making*.

Sincerely,

Marc Ellsworth, MD